Author's response to reviews

Title: Long term evolution of valsalva retinopathy: a case series

Authors:

Miriam Garcia Fernandez (migarci@hotmail.es)
Joaquín Castro Navarro (jcastro22557@gmail.com)
Carmen Gonzalez Castaño (cgcastano@hotmail.com)

Version: 4 Date: 7 July 2012

Author's response to reviews: see over
RESPONSES TO REVIEWERS

Dear reviewers:

We have changed the word “long time” by “long term”.
The manuscript has been checked over by a native English speaker, and there have been corrected all grammatical errors.

Many thanks,

The authors